Detailed Responses to reviewer’s comments

MS: 1350824632162186

In Vitro Antibacterial and Antibiotic-potentiation Activities of the Methanol Extracts from Beilschmiedia acuta, Clausena anisata, Newbouldia laevis and Polyscias fulva against Multidrug-Resistant Gram-Negative Bacteria

Simplice B Tankeo, Pierre Tane and Victor Kuete

Dear Editor, Dear Tom, I am now submitting the revised manuscript taking in account your recommendations. Below are the change made; they are indicated in red color in the manuscript.

I have now discussed this case internally with my colleagues, and we feel that it would be appropriate for you to address the concerns raised by the referee regarding the lack of cytotoxicity tests performed as a limitation of your study within your manuscript.

Answer: The lack of cytotoxicity tests has now been stated as a limitation of our study in the conclusion section

During our assessment, a number of typographical errors were also noted in your manuscript, so we would also encourage you to copyedit it before submitting your revisions.

Answer: The manuscript was carefully checked for typological errors

Best regards
Prof. Dr. Victor Kuete